Author's response to reviews

Title: Acute spontaneous neck haematoma in infants: a rare entity

Authors:

Shimin Zhuang (zhuangsm05@163.com)
Jin Ye (yejin_sums@aliyun.com)
Jingjia Li (li-jingjia@hotmail.com)

Version: 2
Date: 13 August 2014

Author's response to reviews:

We have made some changes in the revised manuscript:
1. Including the email addresses of all authors on the title page;
2. Including a 'Consent' section after the Conclusions;
3. Including a statement regarding consent from patients;
4. Including line and page numbers in the revised manuscript.